Author’s response to reviews

Title: Factors associated with first-line antiretroviral treatment failure in adult HIV-positive patients: a case-control study from Ethiopia

Authors:
Yihienew Mequanint Bezabih (myhiienew@gmail.com; yihienew.bezabih@oniris-nantes.fr)
Fekadu Beyene (bsfekadu@yahoo.com)
Woldesellassie Bezabhe (Woldesellassie.Bezabhe@utas.edu.au)

Version: 3 Date: 04 Jun 2019

Author’s response to reviews:

Vicente Estrada/MD, PhD
Editor-in-Chief, BMC Infectious Diseases

Date: June 4th, 2019

Dear Dr. Vicente Estrada,

Re: INFD-D-19-00077
We addressed your comment regarding the availability of supplementary materials. Thank you

Sincerely,
Yihienew Bezabih, MD
Master 2 student
ONIRIS, University of Nantes, France
Editor Comments:
1. The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based.

We have now changed the statement regarding the availability of data and materials section.